Author's response to reviews

Title: The accessory papillary muscle with inferior J-waves--peculiarity or hidden danger?

Authors:

james ker (jker@wol.co.za)
lorraine du toit (lorraine.dutoit@med.up.ac.za)

Version: 2 Date: 20 October 2009

Author's response to reviews:

Dear Sir/Madam

We have revised the following aspects:

1) We clarified why the patient sought medical advive
2) The clinical implications of the case has been clarified
3) The discussion have been expanded by the citing of the only other published case on accessory papillary muscles and electrocardiographic effects